Weekly Summary
West Region SHG
September 04, 2015

Bandon

No Hot Topics to report, just routine fish culture, liberations, and maintenance.

Cole Rivers
Started marking the 5215 StS
Hauled the last stocking of the year for Upper Rogue legals
No near misses or injuries this week

Elk River

No Hot Topics to report, just routine fish culture, liberations, and maintenance.

Leaburg

No Hot Topics to report, just routine fish culture, liberations, and maintenance.

Marion Forks

No Hot Topics to report, just routine fish culture, liberations, and maintenance.

McKenzie

No Hot Topics to report, just routine fish culture, liberations, and maintenance.

Hatchery has been without internet connection all week.

Minto
Species Weekly Cum. Recycled/outplanted Rel.
Total Total Upstream
STW 0 186 0 186
StS 7 261 298 0
CHS 501 3594 0 1123

Adult collection has picked this week, USACE will start lowering temp. to below 60 this

week.

Crew spend numerous hours stocking and maintaining the display tank at state fair.




Roaring River
From Aug. 29th through Sept. 4th we have received 1.7” of rainfall. The rain was very
welcome as we were stretching the available water as far as it could go. We rely on 100%
reuse water this time of year for 147,000 of our Rb and StS. We have been able to increase
flow to ponds that needed it, and are hopeful that the river level will not drop back to its
previous level.
We are continuing to treat the 2415 and 3315 StS for Ich. The treatments are going well but
the fish are still flashing, we haven’t seen any loss from the Ich. but the fish are not acting
normally. We are optimistic that the treatment will be effective.
Our lib. truck had one tire replaced by Les Schwab due to being really worn and weather
checked. The tire had been on the truck since 1997.
We put together a bulk order of Smith Root electro. fish handling gloves. All together
between RR, Cole Rivers, Trask, and the Mid. Coast STEP program we ordered 11 pairs and
received a 15% discount.
We have been busy painting buildings and preparing spawning equip. as time allows.
We took a load of non-chlorinated ice to the State Fair to bring down the temp. of the ODFW
aquarium due to a fish dying and clogging the chiller inlet. The fish had been removed and
the chillers were operating, but the temperature had gone up to 79 degrees during the time the
chillers were off. The temp. was down to 66 when we arrived, and the remaining fish looked
0.k.

Rock Creek
No Hot Topics to report, just routine fish culture, liberations, and maintenance.

S. Santiam

The Foster Trap was operated on Tuesday of this week. All unmarked ChS are now being
held until water temps drop.

Species Weekly Cum. Recycled* Rel.
Total Total Upstream

Stw 0 129 Na 129

(2415)

StS 2 469 51 Na

(2416)

ChS 80 6220 2437 445

*StS recycled numbers are cumulative and include reruns.

DEQ samples were taken this week.

We are preparing for our first CHS spawn next week.

General fish culture duties, end of the month reports and maintenance rounded out the week.

Willamette
We went to Dexter this week and got another 300 females. Fish are starting to ripen up so we
are going to spawn next week.
Rainbow releases for the week: Faraday and N.F. Res.
All counts were taken and monthly paperwork done.
Sarah was here this week to look over our fish. She found a couple Ich on the smaller STS.
Another parasite we haven’t seen here in a long time if ever. We treated them and will
monitor.



